Author's response to reviews

Title: Paired measurement of urinary creatinine in neonates based on a Jaffe and an enzymatic IDMS-traceable assay

Authors:

Karel Allegaert (karel.allegaert@uzleuven.be)
Pieter Vermeersch (pieter.vermeersch@uzleuven.be)
Anne Smits (anne.smits@uzleuven.be)
Djalila Mekahli (djalila.mekahli@uzleuven.be)
Elena Levtchenko (elena.levtchenko@uzleuven.be)
Steven Pauwels (steven.pauwels@uzleuven.be)

Version: 4 Date: 27 March 2014

Author's response to reviews:

dear editor,

we thank you for the overall positive evaluation of the revised version. As requested, the e-mail addresses of all the authors has been added, a section heading for the conclusions has been added, and the competing interests has been entered just before the acknowledgement section.

we hereby assume that we have answered the additional editorial questions, but we remain available for any further adaptations if felt appropriate.

yours sincerely, also on behalf of all the authors involved,

karel allegaert